Author's response to reviews

Title: Twist and Snai1 expression in pharyngeal squamous cell carcinoma stroma is related to cancer progression

Authors:

Anna Jouppila-Mättö (anna.jouppila-matto@kuh.fi)
Mervi Närkiö-Mäkelä (mervi.narkio-makela@kuh.fi)
Ylermi Soini (ylermi.soini@uef.fi)
Matti Pukkila (matti.pukkila@kuh.fi)
Reijo Sironen (Reijo.Sironen@uef.fi)
Hanna Tuhkanen (hanna.tuhkanen@uef.fi)
Arto Mannermaa (arto.mannermaa@uef.fi)
Veli-Matti Kosma (veli-matti.kosma@uef.fi)

Version: 3 Date: 3 August 2011

Author's response to reviews: see over
3rd August, 2011

Dear Editor

We are delighted to hear that our manuscript entitled "TWIST AND SNAI1 EXPRESSION IN PHARYNGEAL SQUAMOUS CELL CARCINOMA STROMA IS RELATED TO CANCER PROGRESSION" by Jouppila-Mättö et al (BMC Cancer, MS 5885175965250902) is now accepted for publication. We have made the following corrections to our manuscript:

- The reference list is corrected so that all the authors are written up to 30 names.
- We have removed all the highlightnings.
- The “Material and Methods” is corrected to “Methods”.
- The typographical corrections are made as Reviewer Reis proposed.

Thank you for co-operation and good advices.

Sincerely yours,

Anna Jouppila-Mättö, M.D
Department of Otorhinolaryngology - Head and Neck Surgery, Kuopio University Hospital, P.O. Box 1777, FI-70211 Kuopio Finland.
Telephone: +358-17-172820
Fax: +358-17-174920.
e-mail: anna.jouppila-matto@kuh.fi

Veli-Matti Kosma, M.D., Ph.D.,
Institute of Clinical Medicine, Pathology and Forensic Medicine, University of Eastern Finland, P.O. Box 1627, FI-70211 Kuopio Finland.